Author’s response to reviews

Title: Trial design: Computer guided normal-low versus normal-high potassium control in critically ill patients: Rationale of the GRIP-COMPASS study.

Authors:

Miriam Hoekstra (m.hoekstra@thorax.umcg.nl)
Mathijs Vogelzang (m.vogelzang@thorax.umcg.nl)
Iwan C.C. van der Horst (i.van.der.horst@thorax.umcg.nl)
Annemiek Oude Lansink (a.oudelansink@icv.umcg.nl)
Joost M.M. van der Maaten (j.m.m.van.der.maaten@thorax.umcg.nl)
Farouq Ismael (f.ismael@icv.umcg.nl)
Felix Zijlstra (f.zijlstra@thorax.umcg.nl)
Maarten W.N. Nijsten (m.w.n.nijsten@icv.umcg.nl)

Version: 3 Date: 22 December 2010

Author’s response to reviews:

Author list: Miriam Hoekstra, Mathijs Vogelzang, Iwan C.C. van der Horst, Annemiek Oude Lansink, Joost M.M. van der Maaten, Farouq Ismael, Felix Zijlstra and Maarten W.N. Nijsten
Title: Trial design: Computer guided normal-low versus normal-high potassium control in critically ill patients: Rationale of the GRIP-COMPASS study.
Journal: BMC Anesthesiology
MS ID: 1970315924403756

Groningen, 22 December 2010

Dear editor,

Hereby I submit the revised version of our manuscript "Trial design: Computer guided normal-low versus normal-high potassium control in critically ill patients: Rationale of the GRIP-COMPASS study".

We hope that it is ready for publication in its current form. The payment will be made on December 23.

If there are any questions left, please do not hesitate to contact,

Best wishes,

Miriam Hoekstra
Department of Anesthesiology
University Medical Center Groningen